Author’s response to reviews

Title: Cerebral disorders in the first 7 years of life in children born post-term: A cohort study

Authors:

Anne Rolschau (ahrolschau@gmail.com)
Annette Olesen (Annette.Wind.Olesen2@rsyd.dk)
Carsten Obel (co@ph.au.dk)
Jørn Olsen (jo@ph.au.dk)
Chunsen Wu (cwu@health.sdu.dk)
Poul-Erik Kofoed (poul.erik.kofoed@rsyd.dk)

Version: 3 Date: 26 Jan 2020

Author’s response to reviews:

Kolding, 26th January 2020.

Department of Paediatrics
Kolding Hospital
6000 Kolding
Denmark

Editorial Office
BMC Pediatrics

Dear Sir

Thank you very much for the feedback on the paper: “Cerebral disorders in the first 7 years of life in children born post-term: A cohort study”.


Below our response to the issues raised:

1. We notice that author CO is missing from the authors' contributions section.

We are sorry for presenting the contribution section unclear. We have now specified that the term “all” also includes JO. He has thus contributed to interpreting the results and in writing the final manuscript.

2. In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

The participants gave written consent which has now been added to the ethics section.

Yours sincerely

Poul-Erik Kofoed